Author’s response to reviews

Title: Morphologic analysis of the proximal tibia after open wedge high tibial osteotomy for proper plate fitting

Authors:
Oui Sik Yoo (osyoo@corentec.com)
Yong Seuk Lee (smcos1@hanmail.net)
Myung Chul Lee (smcos1@naver.com)
Jae Hong Park (ParkJH@hanmail.net)
Jae Won Kim (JKKIM@corentec.com)
Doo Hoon Sun (smcos1@snu.ac.kr)

Version: 3 Date: 21 Sep 2016

Author’s response to reviews:

This is the revision of “BMSD-D-16-00455R2, Morphologic analysis of the proximal tibia after open wedge high tibial osteotomy for proper plate fitting

Oui Sik Yoo, MS; Yong Seuk Lee; Myung Chul Lee, MD, PhD; Jae Hong Park, MD; Jae Won Kim, MS; Doo Hoon Sun, BMC Musculoskeletal Disorders”

Thank you for your kind and detailed edition. I would like thank you again very greatly for your comments and contributions on making this a better article. We have changed our article as you suggested. We have highlighted the changes in red and have added a detailed description on the areas that you have suggested.

Dear Professor Lee,

Thank you for submitting the revised version of your manuscript "Morphologic analysis of the proximal tibia after open wedge high tibial osteotomy for proper plate fitting" (BMSD-D-16-00455R2). It has been re-assessed by an Editorial Board member and I am pleased to inform you...
that your manuscript has been provisionally recommended for acceptance in BMC Musculoskeletal Disorders. However, before we can pass your manuscript on to our production team, there are a couple of very minor stylistic points that need to be addressed as outlined below:

1) Ethical approval and consent to participate: please provide the exact name of the ethics committee that approved your study (please do not use 'local ethics committee' as the full name of the committee is required)

   New line 316; we edited as you commented.

2) Consent to publish: as there is no identifiable patient data provided in the paper, please simply state 'not applicable' under this heading.

   New line 318, we edited as you commented.

Best regards

Sincerely yours